Fri Apr 21 16:41:56 GMI 2006

Year: 06 Doy: 111
Qobserver: yasukawa
WEATHER COMMVENT: Fri Apr 21 16:42:00 GMr 2006
In extensive, thick cirrostratus overcast, Swind 5 nph, tenp 35 F
end
**pi cs_PROCESSI NG PROBLEMF*: Fri Apr 21 17:19:00 GMVI 2006
Net wor k connecti on went down briefly only on kei ki and kenobi
Mssrestore got disrupted. Restarted nesrestore once network connection
cane back up.
___end___
WEATHER COMMVENT: Fri Apr 21 17:55:11 GMI 2006
Crrostratus thinning, opened domes.

end
COMMENT: Fri Apr 21 17:55:36 GMI' 2006
CGui der left on overnight.

end
Fri Apr 21 17:57:53 GMI 2006 CH P Start Patro
Fri Apr 21 17:58:58 GMI 2006 Pl CS Start Patro

WEATHER COWMMENT: Fri Apr 21 17:59:35 GMI 2006
Started up but patch of thick stuff noved over again, guiding may be iffy.

end
Fri Apr 21 18:00:16 GMI 2006 Pl CS Fl at
Fri Apr 21 18:01:51 GMr 2006 CH P LSD
Fri Apr 21 18:02:27 GJI 2006 Pl CS End Fl at
Fri Apr 21 18:02:32 GV 2006 Pl CS ReStart Patro
Fri Apr 21 18:03:33 GJI 2006 CH P End LSD
Fri Apr 21 18:03:39 GMVI 2006 CH P Bi asLSD
Fri Apr 21 18:04:18 GMI 2006 CH P End Bi asLSD
Fri Apr 21 18:04:24 GMIr 2006 CH P Bi as
Fri Apr 21 18:05:04 GJI 2006 CH P End Bi as
Fri Apr 21 18:05:09 GJI 2006 CH P ReStart Patro
PSPT COWMENT: Fri Apr 21 18:08:25 GMI 2006
observi ng
end

**ML.SO PROBLEM *: Fri Apr 21 18:10: 03 GMI 2006

Declination appears to be drifting. Signal on guider nmonitor is dropping
down from usual mddle position. PICS image, though, remains centered
and stable at this tine.

end

**M.SO PROBLEM *: Fri Apr 21 18:24:04 GMI 2006

Hoi ke images indicating declination drift.

end

**PSPT PROBLEM *: Fri Apr 21 18:28:55 GMI' 2006

PSPT crashed at 1817 UT. K/r




end
**MLSO PROBLEMF*: Fri Apr 21 18:31:08 GMI 2006
Gui der nonitor o' scope dot slowly moving back to nom nal position

Guider still not pointing well but that also could be due to the very thick
clouds I'min at this tinme.

___end

Fri Apr 21 18:55:48 GMVI 2006 CH P End Patro

WEATHER COMMVENT: Fri Apr 21 18:55:58 GMI' 2006
St oppi ng due to thick overcast.

end
Fri Apr 21 18:56:10 GMIr 2006 Pl CS End Patro
Fri Apr 21 20:41:40 GMr 2006 CH P Start Patro
Fri Apr 21 20:41:51 GVIr 2006 Pl CS Start Patro

**M.SO PROBLEM *: Fri Apr 21 20:35:52 GMI' 2006

I n Septenber 1991, the gearing in the declination Vernitron notor failed,
producing a simlar |azy-to-no response in the guiding declination

At that time the notor was replaced with a spare and the old nmotor was sent
back to Boul der for repair

Thi s notor problemoccurred again in May 1995 and in January 2001.

There is a good tip on renmoving the notor in the 01d022 observer | og.

A refurbished motor is currently in the spares inventory.

| also wired up an interconnect to nmake the notor replacenment |ess
contortional

Opened spar and checked gearing between notor assenbly and dec drive

worm gear. GCears feel OK, albeit a bit tight. Renoved notor and checked it.
The notor al so appears OK, spinning without resistance with no | oad.
Reassenbl ed dec drive nmechani sm and took sone photos of it and the PICS

Cl assenbly while it was exposed for photo-docunmentation effort.

end
**MLSO PROBLEMF*: Fri Apr 21 20:49:04 GMr 2006
Al t hough sun is still behind noderate to thick cirrostratus and al tocumul us,
the guiding still appears to drift -- based on o' scope trace.

end

**MLSO PROBLEMF*: Fri Apr 21 20:58:29 GMIr 2006
Declinati on worm screw was closer to one limt than the other. Tried pushing
spar in opposite direction and reset the guider and dec to see if | could
drive dec closer to center and see if it behaves any better over there.
Declination did not drive spar toward centering screw nor the inmage over the
occulter. 1t looks like the electronics may not be providing any drive
voltage to the dec nmotor and the drift we are seeing is just static
drift of the sun’s path relative to the static dec position of the spar

end
WEATHER COMMVENT: Fri Apr 21 21:32:28 GMI 2006
In thick orographic clouds.

end
Fri Apr 21 21:32:48 GJI 2006 CHP End Patro
Fri Apr 21 21:33:16 GJI 2006 Pl CS End Patro

**MLSO PROBLEM *: Fri Apr 21 23:52:12 GMI' 2006



Checked gui der notor voltages at MS connector on spar. 115 VACis K
When dec is reset, pins Cto Dis 26 VAC. Wuen lint switch is pressed,
Cto D goes to 36 VAC (should be 39 VAC). When center switch is pressed,
Cto Dgoes to 0 VAC (dec relay is released). Wen dec is reset again,
Cto D goes to 26 VAC

Vol tages to notor/tach appear to behave properly except that the recenter
voltage is a bit |ow

___end___

WEATHER COMMENT: Sat Apr 22 00:08:13 GMI 2006

In fog, closing domes.

end__

**MLSO PROBLEMF*: Sat Apr 22 01:17:17 GMIr 2006

Revisited the notor and gear train. Softened up the worm gear grease a bit
with a shot of Wb-40. (could not find the grease gun) No change in
problem Checked notor operation under no |l oad with declination power on
and it appeared to run OK at the different voltages (run and recenter).

| could stop the output gear with ny finger at either voltage. Checking
it against the spare notor/tach in the cabinet, both notors behaved
simlarly so | installed the spare. The spare would not run the
declination gear train. | could feel it vibrating (running internally).
A qui ck shot of WD-40 on the bearings of the gearbox between the
notor/tach and the worm gear box did the trick, the notor started turning the
gear train and the declination system began noving as it shoul d.

Tested the limt switch behavior and the dec recentered itself and

turned itself off.

Think it is working now We will have to check the bearings

in the internmedi ate gearbox for wear and the overall lubrication

regreasing the wormdrive with some real, fresh grease sonetime soon
end

NI CE CHI P | MAGE: 1814

Nl CE PI CSLI MB | MAGE: 1824
NI CE PI CSDI SC | MAGE: 1814
WEATHER COMMENT: Sat Apr 22 01:45:49 GMI 2006
Still in thick overcast.
__end____
COWMENT: Sat Apr 22 02:17:09 GMI 2006
10 man-hours on site.
___end___
Sat Apr 22 02:17:48 GV 2006
M1 V



